Title: Traditional Eye Medicine Use by Newly Presenting Ophthalmic Patients to a Teaching Hospital in South-eastern Nigeria: Socio-demographic and Clinical Correlates

Authors:
Boniface Ikenna Eze (xy3165767@yahoo.com)
Chimdi Memnofu Chuka-Okosa (chimdi_8691@yahoo.com)
Judith Nkechi Uche (amazingnkechi@yahoo.com)

Version: 5 Date: 29 September 2009

Author's response to reviews:

Revisions to my manuscript as per the Editor's recommendations
1. I have deleted the words "title page" from the title page of the manuscript.
2. The abstract has been included in the second page.
3. I have enclosed reference no 16 in square brackets thus [16].